Author’s response to reviews

Title: Diagnostic Proficiency and Reporting of Lassa fever by Nigerian physicians in Osun State of Nigeria

Authors:

Samuel A Olowookere (sanuolowookere@yahoo.com)
Akinola A Fatiregun (akinfati@yahoo.com)
Olalere O Gbolahan (gbolahanlere@yahoo.com)
Ebenezer G Adepoju (nezerman@yahoo.co.uk)

Version: 5 Date: 17 June 2014

Author’s response to reviews: see over
The Editor,

BMC Infectious Diseases

Dear Editor,

Re: Submission of the revised manuscript title “Diagnostic Proficiency and Reporting of Lassa fever by Nigerian physicians in Osun State of Nigeria” Manuscript Number: MS: 8315240541174246

Please, find the revised version of our manuscript previously submitted, titled above.

We hereby expressed our appreciation to the editor and the reviewers for the painstaking and constructive review of the article as well as the opportunity given us to revise the manuscript.

The required changes had been made to the manuscript.

Thank you

Yours sincerely,

Dr SA Olowookere

Corresponding author